REGENTS REVIEW US HISTORY –FOR JANUARY 2020

TOPIC	DAY/DATE	TIME	TEACHER/ROOM#
Colonial Times Causes of the Revolution Effects of the Revolution Articles of Confederation US Constitution New Nation	THURSDAY, DEC. 12, 2019	2:30 - 3:30	MR. TROFFA/ RM 214
Expansion of democracy/Age of Jackson 19 th C reform movements Industrialization Manifest Destiny Causes of the Civil War Civil War and Reconstruction	THURSDAY, DEC. 19, 2019	2:30 - 4:00	MR. TROFFA/ RM 214
Gilded Age (Immigration, Urbanization, and Industrialization) Progressive reforms American expansion and imperialism World War I	TUESDAY, JAN. 7, 2020	2:30 - 4:00	MR. TROFFA/ RM 214
1920s and Great Depression World War II Cold War Post-Cold War foreign policy DBQ essay writing	THURSDAY, JAN. 9, 2020	2:30 - 4:00	MR. TROFFA/ RM 214
Civil rights movements 20 th Century American politics Current issues in American politics and society Writing the thematic essay	THURSDAY, JAN. 16, 2020	2:30 - 4:00	MR. TROFFA/ RM 214

US HISTORY REGENTS WILL BE ON WEDNESDAY, JANUARY 22, 2020 AT 8:15AM